Author's response to reviews

Title: Fatal invasive cervical cancer secondary to untreated cervical dysplasia: a case report

Authors:

  Stepan Braun (s.braun@amgen.com)
  Daniel Reimer (daniel.reimer@i-med.ac.at)
  Isolde Strobl (isolde.strobl@i-med.ac.at)
  Ulrike Wieland (ulrike.wieland@uni-koeln.de)
  Petra Wiesbauer (petra.wiesbauer@uki.at)
  Elisabeth Müller-Holzner (elisabeth.mueller@i-med.ac.at)
  Siegfried Fessler (siegfried.fessler@uki.at)
  Arthur Scherer (gyn@sb-brixen.it)
  Christian Marth (christian.marth@uki.at)
  Alain G Zeimet (alain.zeimet@i-med.ac.at)

Version: 4 Date: 11 March 2011

Author's response to reviews: see over
Dear Sirs!

As suggested by the editorial office we provided the term “Caucasian” to the manuscript at page 2, line 6 and at page 5, line 2.

We hope we have been able to address all the Editor’s comments and would be grateful for the revised manuscript to now be considered worthy of publication in JMCR.

Yours sincerely,

Alain G. Zeimet & Daniel Reimer